Author's response to reviews

Title: Solid Variant of Aneurysmal Bone Cyst of the Thoracic Spine: Case Report

Authors:

George Al-shamy (shamy@bcm.edu)
Katherine Relyea (kxrelyea@texaschildrenshospital.org)
Adekunle Adesina (amadesin@texaschildrenshospital.org)
William E Whitehead (wewhiteh@texaschildrenshospital.org)
Daniel J Curry (djcurry@texaschildrenshospital.org)
Thomas G Luerssen (tgluerss@texaschildrenshospital.org)
Andrew Jea (ajea@bcm.edu)

Version: 4 Date: 14 March 2011

Author's response to reviews: see over
Dear Editor,

Thank you for your comments regarding our paper, “Solid Variant of Aneurysmal Bone Cyst of the Thoracic Spine: Case Report.” We have made the following suggested changes to our paper:

1) We have added arrows to Figures 1-4 to show the pathological areas of interest.

2) We have changed the figure labels in the manuscript to better correspond with our figure legends on page 12.

Please let us know if there is anything else we need to do. Once again, we thank you for reviewing our article.

Sincerely,

Andrew Jea MD

Corresponding Author